Author's response to reviews

Title: Long-term symptoms in dizzy patients examined in a university clinic

Authors:

Kjersti Wilhelmsen (Kjersti.Wilhelmsen@hib.no)
Anne Elisabeth Ljunggren (Elisabeth.Ljunggren@isf.uib.no)
Frederik Goplen (Frederik.Goplen@ore.uib.no)
Geir Egil Eide (Geir.Egil.Eide@Haukeland.No)
Stein Helge G Nordahl (Stein.Nordahl@ore.uib.no)

Version: 4 Date: 13 May 2009

Author's response to reviews: see over
Changes have been made as requested by the Editorial Production Team

References 19 and 32 have been updated.

All the tables have been adjusted.

Citation 30 and 31 in the text has changed positions, and on the list of references accordingly.

The author list has been checked and updated.

The statement reading “these authors contributed …..” has been removed and also the * referring to this statement from the list of authors.

Key-words are removed in abstract as part of the article, but retained in the abstract submitted separately.

“List of abbreviation” has been changed to “Abbreviations”

Other changes

Text in the article have been checked and altered with respect to language. Four words have been re-introduced in the abstract (interquartile range, standard deviation).

I hope the changes are according to requirements.

Bergen 13.05.09

On behalf of the authors

Kjersti Wilhelmsen